Author’s response to reviews

Title: ADENOSINE DEAMINASE ACTIVITY IN TYPE 2 DIABETES MELLITUS. DOES IT HAVE ANY ROLE?

Authors:
Apeksha Niraula (apekshyaniraula@gmail.com)
Saroj Thapa (dr.sarojthapa@gmail.com)
Saroj Kunwar (saroj.kunwar00@gmail.com)
Madhab Lamsal (madhab.lamsal@bpkihs.edu)
Nirmal Baral (nirmal.baral@bpkihs.edu)
Robin Maskey (dr.maskey@gmail.com)

Version: 2 Date: 15 Jun 2018

Author’s response to reviews:

1) iThenticate: I have tried to overcome the textual overlap by modifying the text in my own words.

2) English language: I have taken the help from the professional for the english language correction for the article.

3) Authorship: I have incorporated Prof. Nirmal Baral in the submission system as well. I apologize for the mistake.

4) Title page: I have removed the supplementary file version of the title page. I have removed the ‘Disclosure of Interest’ statement on the title page. The author email addresses are consistent between the title page and the submission system.

5) Abstract sections
‘Objective’ subheading has been removed and incorporated in the text into the Background section. ‘Materials and Methods’ subheading has been changed to ‘Methods’.

6) Keywords
4 keywords have been included.

7) Main article format
‘Conclusions’ subheading has been included before the concluding paragraph.
8) Abbreviations
Abbreviations has been incorporated.

9) Consent for publication
Not applicable

10) Availability of data and materials: Not applicable

11)Competing interests: Sentence has been modified accordingly.

12) Authors’ contributions: Sentence modified accordingly.

13) Acknowledgements: Sentence rephrased.

14) Figure legends: Not applicable as there is no figure in the article.

15) Tables: Amendment done as instructed.

Reviewer 2:
1-in the manuscript (abstract and discussion) need to write in ( units) for ADA: Units written in both the section

2-In discussion the word of Hba1c must be write Hb A1c : Done accordingly.